Author's response to reviews

**Title:** Multimodal neuroimaging of frontal white matter microstructure in early phase schizophrenia: The impact of early adolescent cannabis use (study protocol)

**Authors:**

- Denise Bernier (dcbernie@dal.ca)
- Jacob Cookey (jc533393@dal.ca)
- David McAllindon (dmcallindon@gmail.com)
- Robert Bartha (rbartha@robarts.ca)
- Christopher C Hanstock (chris.hanstock@ualberta.ca)
- Aaron J Newman (aaron.newman@dal.ca)
- Sherry H Stewart (sherry.stewart@dal.ca)
- Candice Crocker (candice.crocker@dal.ca)
- Kimberley P Good (kimpgood@dal.ca)
- Philip G Tibbo (phil.tibbo@cdha.nshealth.ca)

**Version:** 2 **Date:** 23 May 2013

**Author's response to reviews:** see over
May 22, 2013

Mr. Carlo Rye Chua
BMC-series Journals
BioMed Central
Floor 6, 236 Gray's Inn Road
London WC1X 8HB

RE: The manuscript  Multimodal neuroimaging of frontal white matter microstructure in early phase schizophrenia: The impact of early adolescent cannabis use (study protocol)

Dear Mr. Chua,

I have received your request for more explicit statements in regards to ethics approval and funding sources for the above-mentioned study. These changes have now been made in the text, pages 18 and 19.

Yours sincerely,

Denise Bernier